Author's response to reviews

Title: An association between seeing incidents of alcohol or drug use in films and young Scottish adults’ own alcohol or drug use: cross sectional study

Authors:

Kate Hunt (kate@sphsu.mrc.ac.uk)
Helen Sweeting (helen@sphsu.mrc.ac.uk)
James Sargant (James.D.Sargent@Hitchcock.ORG)
Heather Lewars (h_lewars@hotmail.com)
Robert Young (robert@sphsu.mrc.ac.uk)
Patrick West (patbwest@gmail.com)

Version: 5 Date: 18 April 2011

Author's response to reviews:

All the requested changes have been made:
- Abstract is now labelled 'Abstract' rather than 'Summary'
- Conclusions added
- Authors' contributions added
- Requested changes to references and in-text citations made
- 'Insert table here' motes removed.
- Highlighting/track changes removed.
- Manuscript checked for typographical errors.

Thank you, and best wishes, Kate Hunt